Author's response to reviews

Title: Effects of Pistachios on Body Weight and Risk Factors of Metabolic Syndrome

Authors:

Xin Wang (xiangpian11@163.com)
Zhaoping Li (zli@mednet.ucla.edu)
Yanjun Liu (dama0ermaosanmao@yahoo.com.cn)
Xiaofeng Lv (neifenmike@126.com)
Wenying Yang (ywy_1010@yahoo.com.cn)

Version: 5 Date: 19 March 2012

Author's response to reviews: see over
Reviewer's report

Title: Effects of Pistachios on Body Weight and Risk Factors of Metabolic Syndrome

Version: 4 Date: 29 February 2012
Reviewer: Sarah Gebauer

Reviewer's report:

Minor Essential Revisions
- In the methods sections, state how the liver enzymes were measured.

Response: description is added as recommended. Line 65-67, page 5.